Author's response to reviews

Title: The prospective application of a hypoxic radiosensitizer, doranidazole (PR-350) to rat intracranial glioblastoma with blood brain barrier disruption

Authors:

Hironobu Yasui (yassan@vetmed.hokudai.ac.jp)
Taketoshi Asanuma (asanuma@cc.miyazaki-u.ac.jp)
Junichi Kino (radiimu@vetmed.hokudai.ac.jp)
Tohru Yamamori (yamamorit@vetmed.hokudai.ac.jp)
Shunsuke Meike (meike@vetmed.hokudai.ac.jp)
Masaki Nagane (nagane@vetmed.hokudai.ac.jp)
Nobuo Kubota (n-kubota@pola-pharma.co.jp)
Mikinori Kuwabara (kuwabara@vetmed.hokudai.ac.jp)
Osamu Inanami (inanami@vetmed.hokudai.ac.jp)

Version: 2 Date: 4 December 2012

Author's response to reviews: see over